Distribution Agreement

In presenting this thesis or dissertation as a partial fulfillment of the requirements for an advanced degree from Emory University, I hereby grant to Emory University and its agents the non-exclusive license to archive, make accessible, and display my thesis or dissertation in whole or in part in all forms of media, now or hereafter known, including display on the world wide web. I understand that I may select some access restrictions as part of the online submission of this thesis or dissertation. I retain all ownership rights to the copyright of the thesis or dissertation. I also retain the right to use in future works (such as articles or books) all or part of this thesis or dissertation.

Signature:

Chibuzor Babalola

Date

Assessing the setting-specific individual and joint roles of water contact and sanitation practices on the prevalence of *Schistosoma haematobium* prevalence in Kano, Nigeria

By

Chibuzor M. Babalola

Master of Public Health Hubert Department of Global Health

> Dr. Jessica K. Fairley Committee Chair

Assessing the setting-specific individual and joint roles of water contact and sanitation practices on the prevalence of *Schistosoma haematobium* prevalence in Kano, Nigeria

By

Chibuzor M. Babalola

Bachelor of Medicine & Surgery (MBBS) University of Ibadan, Nigeria 2014

Thesis Committee Chair: Dr. Jessica K. Fairley, MD, MPH

An abstract of A thesis submitted to the Faculty of the Rollins School of Public Health of Emory University in partial fulfillment of the requirements for the degree of Master of Public Health in Department of Global Health 2020

ABSTRACT

Assessing the setting-specific individual and joint roles of water contact and sanitation practices on the prevalence of *Schistosoma haematobium* prevalence in Kano, Nigeria

By

Chibuzor M. Babalola

BACKGROUND: Nigeria has the highest cases of schistosomiasis in sub-Saharan Africa and accounts for a quarter of the global population requiring treatment. Factors influencing transmission are complex, context-specific, and need to be understood for adequately designed control strategies. This thesis examined the individual and combined roles of water contact and sanitation practices on *Schistosoma haematobium* prevalence in Kano, Nigeria.

METHODS: A cross-sectional study was carried out in five endemic Local Government Areas of Kano in June and July 2019. Urine samples from school-aged children (4 – 18 years) were screened for *S. haematobium* and follow up questionnaires were administered to assess participants' awareness, behavior, and practices. The analytic dataset used 272 participants who had valid laboratory data. Multivariate log-binomial models were used to estimate prevalence ratios (PRs) adjusted for participants' age, gender, water source, and farming practices.

RESULTS: The prevalence of urinary schistosomiasis for this population was 37.5%; 33.7% of whom reported unimproved sanitation use and 84.6% engaged in at least one of six water contact activities. The majority of participants were male (n=221, 81.6%), and mean age was 11.2 (SD = 4.1). After adjusting for covariates of interest, any degree of water contact (moderate or high) was significantly associated with prevalence, aPR = 3.94, 95%CI [1.68, 9.25]; there was no significant association between unimproved sanitation use and prevalence, aPR = 0.99, 95 CI [0.72, 1.37]; and the observed joint effect of water contact and unimproved sanitation use was significantly associated with infection prevalence; aPR = 3.91, 95%CI [1.68, 9.10].

CONCLUSION: These findings not only suggest that water contact might be the key limiting factor on *S. haematobium* prevalence for this population when compared to poor sanitation infrastructure alone. The study further highlights complexities in the measurement of WASH constructs within urinary schistosomiasis transmission and provides novel direction for more robust causal research that considers the embedded roles of WASH exposures.

Assessing the setting-specific individual and joint roles of water contact and sanitation practices on the prevalence of *Schistosoma haematobium* prevalence in Kano, Nigeria

By

Chibuzor M. Babalola

Bachelor of Medicine & Surgery (MBBS) University of Ibadan, Nigeria 2014

Thesis Committee Chair: Dr. Jessica K. Fairley, MD, MPH

A thesis submitted to the Faculty of the Rollins School of Public Health of Emory University in partial fulfillment of the requirements for the degree of Master of Public Health in Department of Global Health 2020

ACKNOWLEDGEMENTS

I would like to express my deepest gratitude to Dr Jessica Fairley, my thesis Chair, for her accommodating disposition, relentless support, forbearing guidance, exceptional feedback, and judicious critique of this work. Thank you for ensuring my progress and completion.

I would also like to thank eHEALTH AFRICA and Innohealth diagnostics for their magnanimous contribution to the success of the field research.

Ultimately, I am indebted to my Father in Heaven for unending grace and a sound mind.

I. BACKGROUND	1
RATIONALE	1
PROBLEM STATEMENT	3
PURPOSE STATEMENT	6
RESEARCH QUESTION AND HYPOTHESIS	7
SIGNIFICANCE STATEMENT	7
II. LITERATURE REVIEW	9
INTRODUCTION	g
POSSIBLE AREAS OF INTERVENTION WITHIN THE TRANSMISSION CYCLE	10
WATER SANITATION AND HYGIENE IN S. HAEMATORIUM TRANSMISSION	12
THE IMPORTANCE OF ECO-EPIDEMIOLOGICAL AND SOCIOECOLOGICAL CONTEXT	13
INDERSTANDING THE ROLE OF CONTACT WITH CONTAMINATED WATER	14
UNDERSTANDING THE ROLE OF POOR SANITATION	16
OVERVIEW OF SITUATIONAL ANALYSES IN NIGERIA	18
CONCLUSION	20
III. METHODOLOGY	22
INTRODUCTION	22
STUDY AREA	22
STUDY POPULATION AND SAMPLE	23
DATA COLLECTION	23
S. HAEMATOBIUM DIAGNOSIS	25
ETHICAL CONSIDERATION	25
ТНЕ ДАТА	26
MISSING DATA	26
OUTCOME MEASURE	26
MAIN EXPOSURES	27
CONFOUNDING ASSESSMENT	27
MODELING EFFECTS	28
IV. RESULTS	31
DISTRIBUTION OF OUTCOME AND EXPOSURES	31
PARTICIPANT CHARACTERISTICS	31
ANALYTIC SAMPLE	32

TABLE OF CONTENTS

FINAL MODEL	32
EFFECTS OF INDIVIDUAL EXPOSURES	32
EFFECT OF JOINT EXPOSURE	34
FIGURE AND TABLES	35

V. DISCUSSION

Key Findings	40
Joint Association	40
CONTACT WITH CONTAMINATED WATER	42
Sanitation	44
STRENGTHS AND LIMITATIONS	46
IMPLICATIONS FOR PRACTICE AND RESEARCH	47

40

I. BACKGROUND

Rationale

Schistosomiasis - historically termed bilharzia – is a parasitic disease currently estimated to affect over 200 million people globally, causes mortality in about 100,000, and results in chronic illnesses for over 20 million(Rosenthal, 2020). An infectious disease especially associated with poverty, schistosomiasis is considered as one of the targeted Neglected Tropical Diseases (NTDs) - "a group of diseases that cause substantial illness and affect the world's poorest people"(Center for Global Health . Division of Parasitic & Malaria, 2011, p. 1). Recent estimates show that approximately 90% of people who are infected and in need of treatment reside in sub-Saharan Africa. Over a quarter of this population, about 26%, are Nigerians (WHO, 2019a).

The parasitic species implicated in schistosomiasis are trematode blood flukes that belong to the genus, Schistosoma (Colley, Bustinduy, Secor, & King, 2014). There are 6 main species commonly known to cause disease in humans. Five of these – *Schistosoma mansoni, Schistosoma japonicum, Schistosoma mekongi, Schistosoma intercalatum*, and *Schistosoma guineensis* result in intestinal schistosomiasis; while the sixth, *Schistosoma haematobium*, causes urinary schistosomiasis (Rosenthal, 2020). Beyond relatively fewer instances of an acute febrile illness, the hallmark of the disease caused by all species is chronicity and significant morbidity usually as a result of inflammatory host immune responses to the presence of eggs laid by parasite worms in the body tissues (McManus et al., 2018). In urinary schistosomiasis, typical acute – to – chronic presentations include haematuria (blood in the urine), dysuria (painful urination), anemia, anorexia, weight loss, weakness, fluid in the ureters, fluid in the kidneys, urinary tract infections, chronic kidney disease, and bladder wall carcinoma (Rosenthal, 2020). Urogenital schistosomiasis has also been associated with an increased risk for HIV transmission (Kjetland et al., February 2006). In areas of endemicity, the highest infection prevalence is usually recorded among children, especially school-aged between 4 and 17, and young adults in typically poor communities (WHO, 2019a).





The cycle of schistosome transmission begins when water bodies are contaminated with miracidia which hatch from eggs shed in the feces and urine of infected individuals. Miracidia develop & multiply exponentially in intermediate host freshwater snails which then excrete the infective free-living cercariae that penetrate the human skin upon water contact (Colley et al., 2014). Apparent, therefore, in the transmission pathway and life cycle of schistosomes, is an intricate link with Water, Sanitation, and Hygiene (WASH). Moreover, a peculiarity of schistosomiasis as a water-borne disease is evident in how humans get infected by schistosome cercariae; through skin penetration from contact with rather than ingestion of contaminated water (Braun, Grimes, & Templeton, 2018). Sanitation in WASH refers to the containment of human excreta. Its peculiarity in schistosomiasis can also be appreciated by its relatively indirect contribution to infection; in that water bodies have to be contaminated and fresh-water snails penetrated rather than waste matter directly ingested (Grimes et al., 2015).

Problem Statement

While the role of sanitation might be obvious for intestinal schistosomiasis, understanding its role in urinary schistosomiasis would be rewarding; seemingly a more indirect pathway to measure for *S. haematobium*. Outside of control of the intermediate host snails, WASH public health interventions that target schistosome transmission could concentrate on preventing human contact with contaminated water and/or prevention of water contamination in itself (Evan Secor, 2014).

Many WASH interventions have aimed at demonstrating a reduction of schistosomiasis prevalence by focusing on the provision of safe water sources and have documented a consequent reduction in water contact (Grimes et al., 2015). A

few other interventions, such as one reported in a Ghanaian study (2012), have been created to specifically reduce contamination with contaminated water (Kosinski et al., 2012). However, countries that have accomplished large strides in reducing schistosomiasis prevalence and transmission are places where robust economic development occurred over time. These developments naturally incorporated all aspects of WASH – including safe water & adequate sanitation (Secor & Colley, 2018).

Grimes et al.(2015) highlight the complex role that sanitation plays in their review; portraying it as a perplexing one to determine, not only as a result of disproportionate reproduction ratio of miracidia to cercariae, but also due to the important contribution of the population setting or context in determining its role and that of water contact for schistosome transmission in that setting. They conclude that the roles of water and sanitation pathways in schistosome transmission among a population are usually specific to that setting (Grimes et al., 2015, p. 5).

In the last ten years, there has been increased awareness and intensification of the global commitment towards the control of NTDs generally by donor bodies, national and international health programs. An example of this commitment is communicated in the London Declaration on Neglected Tropical Diseases (2012). A recurring theme, however, is the need for data to inform the nature and sustainability of programs (Development, 2012). The current control strategy championed by the World Health Organization(WHO) in the last WHA resolution (2012) is focused on providing preventive chemotherapy with praziquantel to at-risk populations in endemic regions that have a disease prevalence of 10 – 50% like Nigeria. They recommend the incorporation of additional measures like health education and WASH only after some level of transmission reduction has been achieved (Freeman et al., 2013).

In the time since, it has become increasingly apparent that whether it be chemotherapy alone or a more integrated approach, "emphasis has to be put on laying the groundwork necessary" (Secor & Colley, 2018, p. 1). For endemic areas, there is always a risk of rapid reinfection which renders unsustained MDA counterproductive. International praziquantel supply, albeit increased in the past decade, has not been sufficient to meet the need even for one-time campaigns. (WHO, 2013). Moreover, context-specific data is needed to determine what the minimal WASH levels are needed for a setting. And for this data to be generated, "interventions beyond MDA must become widely available" (Secor & Colley, 2018, p. 1). Provision of safe water and adequate sanitation have been key game changers in successful schistosomiasis control programs and are always options for an integrated approach; one that allows for efficient use of the resources available to achieve a more sustainable schistosomiasis control (M. D. French et al., 2018).

Nigeria, Africa's most populous country, has the largest number of people infected with schistosomiasis – about 26% of all infected sub-Saharan Africans (Dawaki et al., 2016). Although both *S. mansoni & S. haematobium* are prevalent; a recent review documented *S. haematobium* as the most prevalent species in Nigeria. (Ezeh, Onyekwelu, Akinwale, Shan, & Wei, 2019). If "sub-Saharan Africa could lose US \$3.5 billion of economic productivity every year as a result of schistosomiasis" (Tanser, Azongo, Vandormael, Bärnighausen, & Appleton, 2018), then about 26% of this loss is realized in Nigeria alone. The WHA 65.21 resolution named 2018 as the milestone for all top 10 endemic countries to initiate control programs. Nigeria is yet to take up this mantle; most probably due to a lack of more suitable data that could guide the development of a national strategy (WHO, 2013). Praziquantel campaigns targeting schistosomiasis in Nigeria have been sporadic at best, with no improvement in consistency since the last WHA resolution. (Ojurongbe et al., 2014). For Nigeria to tackle schistosomiasis headlong, there is a need for ownership and commitment to what would is sure to be a long undertaking. This would be made feasible by a strong foundation of situational data (M. French & Evans, 2019).

Purpose Statement

This pilot study for this thesis was carried out in five endemic Local Government Areas (LGAs) Kano, Nigeria – a northern state in which both intestinal and urinary schistosomiasis are endemic (Abdulkadir et al., 2017). The population surveyed was school-aged children from ages 4 to 18 years. Nested in primary research to discriminate between tools for optimal urinary schistosomiasis diagnosis, this survey specifically aimed to create a background for much stronger epidemiological assessment of the individual and joint roles that contact with contaminated water and sanitation behaviors play in this population. With the recent robust research into S. mansoni and *S. haematobium* in Kano residents by Dawaki et al (2016), this pilot study ultimately hopes to create traction that draws attention to

this population and aid implementation of adequate schistosomiasis control programs for this setting.

Research Question

What are the individual and combined effects of human water contact and sanitation practices on the prevalence of *Schistosoma haematobium* (Urinary blood fluke) among school-aged children in Kano, Nigeria?

Null Hypothesis: There is no difference between the individual and combined effects of water contact and sanitation practices on the prevalence of *Schistosoma haematobium* for this population.

Significance Statement

A highly diverse and "multi-ethnic country" with 380 languages, Nigeria, more than any other, needs suitably designed programs to tackle myriads of health issues, including schistosomiasis (Adedini, Odimegwu, Bamiwuye, Fadeyibi, & De Wet, 2014). Setting-specific data to guide initiation of, commitment to, and efficient delivery of any health program is necessary for the different affected regions. Several prevalence studies carried out in different states of Nigeria are currently being used to provide statistics as needed. However, the consensus is that Nigeria does need more situational data for the promise of any successful control strategy. There is a call for more targeted data to inform the best use of resources for the Nigerian context; in the event of favorable political will and available funding (Ezeh et al., 2019). It is indisputable that water and sanitation contribute considerably to the transmission of schistosomiasis. It has been argued that the more indirect role of sanitation, and probably some of the complexity in effecting it as a public health intervention might relegate it to the background in the design of WASH programs targeting schistosomiasis (Grimes et al., 2015). The demand for long term commitment in the combat of schistosomiasis necessitates understanding all possible tools and approaches possible for the undertaking. A country like Nigeria with not only the highest cases of schistosomiasis, but also repeatedly unsustained or failed myriad public health interventions can benefit from every foresight (Muhammad, Abdulkareem, & Chowdhury, 2017).

What is an estimate of the role water contact plays in urinary schistosomiasis for this population? What does the estimate for sanitation look like? Is there a combined effect worth noting? Ultimately, one role isn't magnified over the other – water contact over sanitation or vice versa and allocated resources are efficiently used. This pilot investigation aims to provide these estimates for Kano, Nigeria. It hopes to highlight the WASH roles in transmission for this setting and leverage its findings for stronger and more robust causal investigations that a national strategy or programs can run with.

II. LITERATURE REVIEW

Introduction

Urinary or urogenital schistosomiasis alludes to *Schistosoma haematobium's* characteristic invasion of the urinary system, unlike other species that cause intestinal schistosomiasis (Gryseels, Polman, Clerinx, & Kestens, 2006). Nevertheless, like other schistosomes, the hallmark of the infection with *S. haematobium*, usually irrespective of intensity, is characterized by chronicity that culminates in significant morbidity (C. H. King, Sturrock, Kariuki, & Hamburger, 2006). A case burden of around 250 million and an at-risk population of about 779 million people convey how substantial the healthy years of life lost to the disease is – 90% of which is shouldered by sub-Saharan Africa (SSA) and 26% of this 90% by Nigeria (Hajissa et al., 2018; Tanser et al., 2018). In this regard, schistosomiasis has been referred to as "the most important water-based disease"(P. Steinmann, J. Keiser, R. Bos, M. Tanner, & J. Utzinger, 2006).

Unfortunately, the age-old battle against schistosomiasis has neither been easy nor straightforward (Bergquist, 2013). Albeit waterborne, schistosomiasis demonstrates one of its major complexities in that humans are not infected after ingesting contaminated water, but rather after infective worms penetrate the human skin (Grimes et al., 2015). The processes involved in water contamination by miracidia from human excreta, development of miracidia to cercariae in intermediate-host snails, and infection of human hosts with cercariae tend to overlap (Holveck et al., 2007). Ambiguity persists when both urinary and intestinal schistosomiasis are prevalent in a population (M. D. French et al., 2018).

Possible Areas of Intervention Within the Transmission Cycle

In order to combat urinary schistosomiasis, four main areas that could inform the choice of intervention have been identified. I) The major strategy employed for "morbidity control", the worm burden in the human body could be targeted for reduction. The lone medication for this is praziquantel. 2) Targeting intermediate *Bulinus* snail hosts. 3) Preventing contamination of water bodies with the urine of infected human hosts. 4) Preventing contact between contaminated water and human skin (Rollinson et al., 2013).

Morbidity control with praziquantel is the programming approach recommended by the World Health Organization for many endemic countries, especially because it is invariably the quickest and easiest to carry out on a large scale (WHO, 2013). This has generally been implemented as school-based mass drug administrations. However, for many reasons, including the short-sightedness of treating symptoms instead of the source, there has been mounting evidence portraying the need for more integrated and sustainable approaches (C. H. King, 2009).

To name a few challenges documented in SSA, cure rates after drug administration have been demonstrated to be often suboptimal (Doenhoff et al., 2009; N'Goran et al., 2001); Infection levels return to baseline magnitudes after a maximum of two years(Gray et al., 2010); funding has been unsustainable; and there are looming issues of praziquantel resistance (Hotez, Engels, Fenwick, & Savioli, 2010). Research into the few instances of mass drug administration that have taken place in endemic regions within Nigeria has also proven the paucity, inconsistent programming and return-to-baseline infection levels (Chiamah, Ubachukwu, Anorue, & Ebi, 2019; Garba et al., 2013).

Any move toward the second area of intervention, which involves targeting intermediate host snails biologically, chemically or environmentally could arguably be the most rewarding approach (Angelo et al., 2018). This is because the absence of intermediate host snails is nearly equivalent to the absence of the infection. More affluent countries have been able to anchor schistosomiasis elimination on annihilating the implicated snail species (Sokolow et al., 2016). One lesson that stands out from China's 50-year war against *S. japonicum*, a species that causes intestinal schistosomiasis, is that the "snail control" undertaking requires substantial commitment - an intensive and very expensive commitment (Wang, Utzinger, & Zhou, 2008).

Few countries such as Saudi Arabia, Tunisia, Mauritius, and Egypt that have achieved significant control and/or elimination of *S.haematobium* have occasionally in their integrated schistosomiasis programming, supplemented with snail control when feasible (Knopp et al., 2013). Notwithstanding, the bulk of strategies employed in these countries have revolved around development, behavioral modification and political commitment toward local water, sanitation and hygiene structuring (Secor & Colley, 2018).

Water Sanitation and Hygiene in S.haematobium Transmission

In 2010, Bartram and Cairncross began to develop the idea that in spite of being an obvious necessity, Water, Sanitation and Hygiene (WASH) had become one of the "forgotten foundations of health" (p. 1). This points to the characteristic tie of neglected diseases like schistosomiasis to an inherent detriment in WASH systems within any affected population. And in that the disease is neglected, WASH systems are typically found lacking (Charles H. King, 2017). A notable occurrence is that many successes against schistosomiasis have been achieved due to a natural improvement in WASH structures concurrent with economic development while most others have been intentional commitments (Bergquist, 2013).

Egypt, Tunisia, and Algeria stand out in Africa as countries that have made huge strides in the intentional control and/or elimination of *S.haematobium*. In all, minimal progress is being realized in endemic areas of Africa to the south of the Sahara (Rollinson et al., 2013; J. Utzinger et al., 2009). After assessing the nearstagnant cycles of schistosomiasis control in SSA, Stothard, Chitsulo, Kristensen, and Utzinger (2009) emphasize the inescapable need for integration of WASH focused interventions with morbidity control to achieve more sustainable outcomes (p. 1).

For a simple sketch, the "water" component of *S.haematobium* transmission is rooted in water contamination and exposure to contaminated water. Sanitation, which is the containment of human excreta, is also related to water contamination; from direct urination into the water body to more complex mechanisms that might link a lack of sanitation to water contamination (Grimes et al., 2015). Water contamination and contact are both human acts. Therefore, hygiene, which is contingent on human behavior, could very much be the backbone of both water and sanitation (Lansdown et al., 2002). The strength of this behavioral determinant in transmission has become a propellant of the need to understand socio-ecological and eco-epidemiological contexts within which frailties in WASH structures act to determine infection prevalence and intensity so as to tailor behavioral interventions appropriately (J. Russell Stothard, French, Khamis, Basáñez, & Rollinson, 2009).

The Importance of Eco-epidemiological and Socioecological Context

The century-long fight against schistosomiasis is in complete agreement of the fact that the "infection is highly focal, often with substantial heterogeneity between neighboring communities (M. French & Evans, 2019)." Clements et al (2009) were able to apply Bayesian geostatistical prediction tools to evaluate the variations of local *S.haematobium* infection in Burkina Faso, Mali, and the Niger. They provide evidence that local infection or transmission patterns are not always representative of the whole and that variability should be expected within the same geographical location. (pp. 921 - 927).

This setting-specific approach to research and control also extends to praziquantel delivery and snail management. The underpinnings of other major lessons from China's incredible success against *S. japonicum* is found in the significance of adapting strategies to different settings within the country (Wang et al., 2008). Findings from an assessment of drug response and reinfection rates after praziquantel treatment of *S.haematobium* infection in an endemic region of the Ivory

coast supports the reasoning that even a seemingly straightforward approach could require population-specific knowledge (N'Goran et al., 2001). Within integrated control, therefore, is the importance of local context.

Creating WASH solutions for urinary schistosomiasis has likewise been neither clear cut nor seamless (Freeman et al., 2013). Usual directions have involved doing what seemed obvious – providing safe water sources. However, due to several factors, specifically hinged on understanding the setting-specific roles of water contamination and water contact, results have varied for different interventions with varying indicators measuring impact in varying populations (Aagaard-Hansen, Mwanga, & Bruun, 2009).

As the response to the urge for more research into transmission dynamics that pertains to WASH increases, the most basic of questions that arise revolve around determining which factor limits transmission in a population and the extent to which it does (Colley & Secor, 2007; Grimes et al., 2015). Is water contact a sole limiting factor? What role does poor sanitation play in the setting? And how unprofitable might focusing on one over the other be?

Understanding the Role of Contact with Contaminated Water

Human contact with contaminated water is the frame of reference within which both schistosome penetration of the human skin occurs and contamination of water with infected urine is most likely to occur. For this reason, "water contact" has remained a prime determining factor of infection intensity and prevalence (Bruun & Aagaard-Hansen, 2008). Because most endemic regions like those in Nigeria typically lack potable water sources, water contact is almost inevitable, even besides recreational reasons (Espino, Koops, Manderson, Research, & Training in Tropical, 2004). Therefore, research has sought to prove this association in myriad settings. In addition, different water–based interventions have been evaluated to quantify their effects on *S. haematobium* transmission in different settings. What has proved most apparent, outside of unavoidable research limitations, are the intricate complexities of estimating indices for water contact or the factors associated with it and then providing context-appropriate interventions that are impactful enough to prevent it (Aagaard-Hansen et al., 2009).

Many studies in sub-Saharan Africa have been able to prove an association between proximity of an affected population to the prevalence of urinary schistosomiasis (J. Utzinger et al., 2009). Others have linked infection incidence with the development of large water dams or irrigation systems (P. Steinmann et al., 2006). Studies in Kenya that modeled the adjusted odds of infection based on estimated water contact with snail infested water bodies have proven increased odds (Clennon, Mungai, Muchiri, King, & Kitron, 2006). Rudge et al. (2008) assessed high-risk behaviors involved in transmission for a Zanzibar setting. They reported the significance of evaluating water contact patterns as a platform for further estimating behavior (pp. 45 - 53). In some instances, null associations have been found and in others, ambiguity (J. Russell Stothard et al., 2009). One study explored water contact practices among girls in rural South Africa but could not estimate any differential effects between those who reported exposure and those who did not (Thomassen Morgas, Kvalsvig, Taylor, Gundersen, & Kjetland, 2010). Water-based interventions have varied in their impact as much as the type of intervention and settings have varied. It is accepted that the peculiarities of schistosome transmission and behavioral context have manifested as difficulty in reducing water contact by simply providing safer water sources (Grimes et al., 2015). Consequently, some interventions have focused on and recorded a reduction in water contact as a proxy for the reduction in infection burden (Tchuem Tchuenté, Southgate, Webster, De Bont, & Vercruysse, 2001). One Ghanaian intervention was able to measure significant reduction in prevalence of *S.haematobium* after the provision of a safe water recreation area. This is an example of a tailored intervention (Kosinski et al., 2012).

Understanding the Role of Poor Sanitation

Uncertainties surround the role that lack of sanitation plays in transmission and prevalence of urinary schistosomiasis, including how to measure its effect (Asaolu & Ofoezie, 2003). The proportion with which miracidia from water contamination multiplies into cercariae from intermediate host snails is usually exponential, reasoning that might be behind focusing on provision of safe water sources alone. However, this reproductive potential has mainly been measured for intestinal schistosomiasis (Grimes et al., 2015). Furthermore, the effect of sanitation on direct urination into water bodies is vague. While some findings suggest that this gives sanitation a pivotal influence, others simply have not estimated its role (Bruun & Aagaard-Hansen, 2008). Another interesting challenge is that intestinal and urinary schistosomiasis are typically co-endemic within countries like Nigeria and so how proportionate could keeping stool from contaminating water bodies be with preventing urine contamination (Rollinson, 2009)? Also, the urge to urinate is increased in urinary schistosomiasis and reportedly relieved with the cooling effect of immersion in water (Rudge et al., 2008). Besides, the provision of sanitation infrastructure has not always translated into use or sustained use which is due to several interwoven factors that have to be considered in choosing the type of sanitation solution that would be beneficial. This has been a narrative in many sanitation programs within Nigeria (Asaolu & Ofoezie, 2003).

A detailed study in Zanzibar found significantly increased odds of *S*. *haematobium* infection due to the actual act of urinating in water (J. Russell Stothard et al., 2009). An ensuing study found reduced odds of infection among those who had latrines in their homes (Knopp et al., 2013). Some other estimations have been indirect, such as simply demonstrating correlations in areas that lack sanitation (Liao et al., 2011). A review of control and elimination strategies all over the world found a few effects of sanitation improvements alone on prevalence comparable to the provision of safe water sources together with sanitation improvements (Asaolu & Ofoezie, 2003). A post sanitation behavioral intervention in Tanzania was able to achieve a secondary effect of reduction in water contact and a review of WASH improvements with economic developments noted more significant reductions in prevalence when both sanitation access and water sources were improved (Mwanga & Lwambo, 2013).

Overview of Situational Analyses in Nigeria

S.haematobium is documented as the more prevalent species in all regions of Nigeria – Northwest, Northeast, Northcentral, Southwest, and Southeast (Abdulkadir et al., 2017). However, *S.mansoni* tends to exist concurrently in many states, including Kano in the Northwest (Dawaki et al., 2015). Across both species, a 2015 review estimated about 29 million infected people; 55% of whom were children (Adenowo, Oyinloye, Ogunyinka, & Kappo, 2015). The distribution of urinary schistosomiasis is characteristic in that it has mostly been reported among the rural and peri-urban poor who have limited to no potable water sources and improved sanitation. Populations in Northern Nigeria are generally the most representative of this description (Muhammad et al., 2017).

Per the last WHA resolution by the world health organization, Nigeria released a strategy for control, elimination, and strategy in 2012 (WHO, 2013). The strategy highlighted the urgent need for situational analysis and mapping of the disease across all local endemic sites to generate data suitable enough to guide a national control plan. It also took care to specify the need for integrating WASH measures along with the recommended praziquantel chemotherapy as the latter was not sustainable (Onyebuchi, 2012).

Notably, the bulk of research carried out in the past two decades has mainly sought to estimate prevalence and demographics within the settings studied – seeking to generate data that displays endemicity in Nigeria (Ezeh et al., 2019). Studies estimating prevalence levels after praziquantel administration have portrayed this control measure as unreliable (Chiamah et al., 2019; Garba et al., 2013). Some have estimated correlations between *S.haematobium* and other infections involving the urinary tract (Eyong, Ikpeneme, & Ekanem, 2008; Nmorsi, Kwandu, & Ebiaguanye, 2007). Others have focused on mapping out snail populations in some settings (Abe et al., 2018; Akinwale et al., 2011).

Amongst the wealth of prevalence studies are a few investigations into determinants of infection, particularly to establish risk associated with water sources. A 50-year review of urinary Schistosomiasis by Ezeh, Onyekwelu, Akinwale, Shan, and Wei (2019) surveyed 47 out of 323 Nigerian dams and reported that 96% of surveyed dams were located in hyperendemic regions of the country (pp. 1 - 2). One study in southwestern Nigeria was also able to model an association between infection and proximity to a local water source (Ugbomoiko, Ofoezie, Okoye, & Heukelbach, 2010). Eight years after the provision of a safe water source to a community in the South, Ekanem et al. (2017) were able to demonstrate a significant reduction in *S.haematobium* prevalence from 51% to 15% (p. 1).

In Northern Nigeria where Kano state is, research that estimates correlations between water contact activities among school children and prevalence of *S.haematobium* has increased over the past decade (Amuta & Houmsou, 2014; Atalabi, Adoh, & Eze, 2018; Singh, Muddasiru, & Singh, 2016). One of these was able to report the actual increase in odds of infection with water contact activities (Umar et al., 2017). So far in Kano, a relatively comprehensive mapping was carried out between 2005 and 2007 which revealed an *S.haematobium* prevalence range of 18% to 64%. Dawaki et al. (2016), in response to the need for more research that directly estimates risk patterns in Kano, further carried out a hypothesis-generating analysis of some risk factors correlated with combined Schistosoma infections in 5 endemic kano districts (p. 1).

Most studies in Nigeria that have assessed risk factors for *S.haematobium* have mainly reported correlations or prevalence estimates among those who engage in water contact activities. Strengths of association were rarely determined as valid estimates that could inform much-needed programming or more in-depth research. Furthermore, hardly any evaluation of the role of poor sanitation among affected populations is documented.

Conclusion

Urinary Schistosomiasis is a neglected disease and a disease of poverty particularly confined to the rural poor in sub-Saharan Africa who disproportionately lack access to safe water sources and improved sanitation (Freeman et al., 2013; Kullmann et al., 2018). Strikingly, it has been found that "disparities between the rich and poor are much greater in sanitation than in water" (Moe, 2009). Nigeria aptly models this narrative in that only 26% of the population have access to improved sanitation sources and twice of that, some access to safer water sources. What might be more important to note, however, is that populations who lack access to safe water typically lack access to sanitation (Akpabio, 2012).

The "neglect" of diseases like schistosomiasis occurs at both global and national levels in the world (Holveck et al., 2007). This reality, in addition to the level

of integration and commitment required for disease control and elimination, creates an inevitable need for the efficient use of limited resources available sustainably. Nigeria, just like many other endemic nations, is in the race against time and disease complications to develop current and practical guidelines for its different settings (Ezeh et al., 2019).

Within *S. haematobium* transmission, the overlapping intricacies of water contamination, poor sanitation and contact with contaminated water lend credence to a possible need for combined research and combined WASH interventions (C. H. King et al., 2006). It is without reasonable doubt that endemic areas such as Kano Nigeria are not only characteristically proximate to unsafe water bodies but also tend to be areas lacking in sanitation (Jürg Utzinger, N'Goran, Caffrey, & Keiser, 2011).

III. METHODOLOGY

Introduction

This thesis aims to investigate the individual and joint roles of two major exposures - water contact activities and sanitation on the prevalence of urinary schistosomiasis (*S. haematobium*) in Kano, Nigeria.

To this effect, data were taken from a study designed to compare different diagnostic tests for urinary schistosomiasis with a secondary aim to analyze WASH factors associated with infection. This cross-sectional pilot was carried out between June and July of 2019 in select Local Government Areas (LGAs) of Kano.

Study Area

Kano state (11°30'N 8°30'E), located in the central part of Northern Nigeria, is the most populous of the Nigerian states with 13,076,872 people (Portal, 2020). It has an altitude of 400 – 800m above sea level, average annual rainfall of about 800 to 900mm, Sudan Savannah vegetation and widely distributed open water bodies. Its climate is tropical with two predominant seasons – wet (June to September) and dry (October to May). The Hausa and Fulani are the main tribes and they practice agriculture as one of their major occupations (Dawaki et al., 2015).

All of Kano's 44 Local Government Areas (LGAs) were comprehensively mapped between 2005 – 2007 to estimate schistosomiasis prevalence and found to be endemic for *S. haematobium* (Abdullahi, Bassey, & Oyeyi, 2009). Four of the highly endemic LGAs were chosen for this study - Kura with the highest documented

prevalence (62%), Minjibir, Gwarzo, and Bebeji. The kano Municipal area, expected to have relatively lower prevalence, was chosen as a fifth study site (Bassey & Umar, 2004) – Figure 1.

Study Population and Sample

Pre-school and school-aged children between 4 to 18 years of age were the participants from whom urine samples were collected and screened laboratory diagnosis of *S. haematobium*. Follow up questionnaires that mainly assessed a few participant characteristics, WASH - related behavior, disease awareness, and previous praziquantel campaigns were administered to every participant, along with their guardians.

A priori simple random sample size of 275, accounting for a likely 10% loss of data, was estimated based on recently documented prevalence from a survey that had been conducted in three of the LGAs being assessed (Dawaki et al., 2016). Notably, this survey had sampled individuals of all age groups and their reported prevalence of 17.8% used in our sample size calculation was for both *S. haematobium* and *S. mansoni* infections.

The main pediatric General Hospital or Primary Health Center (PHC) in each LGA was chosen as enrollment site.

Data Collection

Data collection was implemented on an on-going basis until the sample size was attained. One extra day of collection was added after attaining 275 urine samples

for one LGA where mild errors in the collection had been noted during iterative process monitoring and evaluation.

Outpatient days specific to each Healthcare center were used as enrollment days. Two research assistants assigned to each LGA used a randomly selected number to determine recruitment within the healthcare facility. For instance, a random number of '3' meant that every third outpatient between the ages of 4 – 18 who came in during the clinic hours would be screened for willingness to participate and enrolled after they or their guardian consented. Every child less than 18 years of age could only be enrolled if they had a guardian with them. Consent and assent forms were signed by the guardian and participants after the study was explained to them. Children less than 6 years of age gave verbal assent.

Following consent and assent, identification numbers were assigned. The participants were instructed on how to pass urine into 50 ml universal sample containers labeled with their identification numbers. Those who could comprehend were asked to pass "terminal urine" if feasible, but otherwise as much urine as they could. The time of urine collection was between 10:00 am and 2:00 pm which is the reported window for optimal *S. haematobium* egg passage (WHO, 1991, pp. 41-44).

After urine samples were obtained, electronic questionnaires built on the Open Data Kit (ODK) platform were administered to participants, with input from their guardians as required. The interviewers asked the questions and input the answers supplied.

S.haematobium Diagnosis

Each viable sample (intact bottle with at least 12 ml of urine) was initially assessed macroscopically and tested for microhematuria with the dipstick test. 10 ml of urine was thereafter filtered, using the standard nucleopore filtration technique. Each filter paper was then stained with Lugol's iodine before the identification of *S. haematobium* eggs under the microscope (WHO, 1991). Diagnosis of *S. haematobium* infection was recorded as the presence or absence of eggs. One trained Research Assistant and a Graduate microbiologist viewed all the samples. An expert microbiologist subsequently viewed 20% of the samples for validation.

Ethical Considerations

The ethical approval for this study was provided by the Institutional Review Board of Emory University (IRB00111157). Secondary permission was also obtained from the Kano State Ministry of Health. This study included a non-invasive collection of urine samples from human subjects, therefore expedited IRB approval was granted on June 18, 2019, after protocol and research instruments were submitted.

Full procedures, potential risks, and benefits were explained to every participant and guardian, after which informed consent and assent were obtained from guardians and participants before data collection. Consent forms were transcribed to the Local Hausa language and back-translated as recommended by the IRB. Every urine sample and questionnaire entry was de-identified; labeled only with the participant IDs. Finally, every child who tested positive for *S.haematobium* was contacted and prescribed a dose of praziquantel through the hospital management.

The Data

Laboratory data recorded and validated by all three lab scientists were doubly entered into excel. The survey data collected electronically on the ODK platform was automatically uploaded to a server and available in excel format, resulting in two major datasets.

Data preparation and subsequent analyses were done using **SAS 9.4** (**SAS Institute, Inc., Cary, NC**). Individual lab and survey excel files were imported into, cleaned and merged to a final analytical dataset. Further coding, cleaning, and manipulation were performed on this final dataset to prepare the variables needed for analyses.

Missing Data

During questionnaire design, most question options included "don't know" and "refused" in addition to "yes" or "no". For consistency across variables, the approach adopted in the manipulation of variables for this thesis was to recode "don't know" and "refused" data values as "missing" (Winters & Netscher, 2016).

Outcome Measure

For this analysis, the outcome of interest was whether *S. haematobium* was present or not. This measure was already specified in this format from the microscopy results.

The Main Exposures

Six *water contact activities* were assessed in the survey: playing, swimming, laundering, working, fetching, and fishing. Participants answered "yes" or "no" to each

of these. In consideration of potential correlation between all these binary variables, a scale was created from 0 – 6; '0' representing participants reporting no water contact and '6' representing participants reporting all water contact activities. From this scale, a measure for "exposure intensity" was created as a variable with three levels – 0, 1 and 2. An intensity of '0' was for a scale of 0; '1' was for a scale of 1 – 3 water contact activities; and '2' was for a scale of 4 – 6 water contact activities. The three levels of intensity were labeled as "none", "moderate/some" and "high" (Table 1).

Sanitation was assessed in this survey by asking for what type of sanitation facility the participants had access to. Participants could choose as many that applied, and the choices ranged from "no facility" to numerous unimproved and improved sanitation types. A dichotomous exposure variable to represent sanitation for analysis was created from this question. This variable was named "unimproved"; coded '1' for those who used unimproved sanitation and '0' for those who did not (Table 1). Unimproved sanitation was created from positive answers to "no facility/open", "bucket", "hanging latrines" and "pit latrine without slab" (WHO/UNICEF, 2019).

Confounding Assessment

Other covariates associated with both outcome and either exposure were considered a *priori* as potential confounders:

- i. Participant' age (Osakunor, Woolhouse, & Mutapi, 2018).
- ii. Participant' gender (Zida et al., 2016).
- iii. If a guardian had formal education or not (Donohue et al., 2017).

- iv. If participant or guardian had heard of schistosomiasis (Folefac et al., 2018).
- v. Participant' involvement in subsistence rice farming (Yapi et al., 2005).
- vi. Unprotected household water source (Hajissa et al., 2018).

The prevalence of urinary schistosomiasis in this study was not rare (> 10%). Therefore, log-binomial regression was used to estimate Prevalence Ratios (PRs) to more accurately reflect the observed relationships between exposures and disease.

Confounding assessment was carried out to inform which variables would remain in the final model as confounders. With the *gold standard* model containing all six potential confounders above, repeated log-binomial regression models were run to assess different combinations of these variables (*sub-models*); when one or more variables were dropped from the *gold standard* model. PRs for individual exposureoutcome associations from each *sub-model* were compared to the estimates from the *gold standard* to show if there was evidence of confounding (a meaningful >10% change in PR) by the dropped variables. The PRs of the chosen model were meaningfully different from that of the *gold standard*. This model contained four confounders – age, sanitation, farming, and water source.

Modeling Effects

Ln (probability of *S. haematobium*) = $\alpha + \beta 1$ (Water Contact) + $\beta 2$ (Sanitation) + $\gamma 1$ (Age) + $\gamma 2$ (Gender) + $\gamma 3$ (Farming) + $\gamma 4$ (Source of Water) + δ_1 (Water Contact * Sanitation)

This analyses modeled the effect of two exposures (water contact and sanitation) on a health outcome (*S. haematobium* infection). The only interaction considered was that between the two exposures of interest. A chunk test was used to

access statistical interaction – as one exposure, intensity of water contact, had three levels and was included in the model using dummy variables. An exposure assessment was thereafter conducted for statistical purposes. However, none of the exposures were dropped from the model, as the research goal was to estimate the effect of multiple exposures. Finally, estimate statements were used to determine crude and adjusted effects.

VARIABLE DESCRIPTION	CATEGORY	COMPOSITION/ DESCRIPTION	CODING	
<i>S. haematobium</i> Infection	OUTCOME	Microscopic diagnosis of S. haematobium	1: present 0: Absent	
Intensity of water contact exposure	EXPOSURE 1	Of the water activities assessed, exposure is high if the participant engages in 4 to 6 activities and moderate for 1 to 3 activities.	2: High exposure 1: Some/Moderate exposure 0: No exposure	
Type of sanitation	EXPOSURE 2	Various types of sanitation facilities	1: Unimproved 0: Improved	
Age	CONFOUNDER		Participants' age	
Gender	CONFOUNDER		1: Male 0: Female	
Source of water	CONFOUNDER	Reported sources of water for everyday use	1: Unprotected 0: Protected	
Farming as a domestic occupation	aing as a estic CONFOUNDER pation		1: Farm 0: Do not farm	

Table1. Variable Specification

Figure1. Local Government Areas



Map Production: eHealth Africa Production Date: 5/20/2019

IV. RESULTS

Distribution of Outcome and Exposures

Table 1 shows the overall prevalence of *S. haematobium* for this study. Of the 272 screened urine samples, 102 (37.5%) participants had the infection.

The water contact activities had a total of 242 valid responses after creating the combined intensity scale (Table 2.1). No water contact was recorded for 81 (33.5%) participants. 100 (41.3%) participants had some water contact, while 61 (25.2%) had a high intensity of water contact. Table 2.2 shows that out of a total of 265 participant responses on sanitation type, 80 (30.2%) used unimproved sanitation facilities. 185 (69.8%) used improved facilities.

Of the participants who were positive for *S. haematobium*, 33.7%(n = 34) reported unimproved sanitation use. 84.6% (n = 77) reported engaging in at least one of playing, swimming, laundering, fishing, fetching, or working in local water bodies (Table 3).

Other Participant Characteristics

Table 3 describes the overall participant characteristics of interest and then stratified by the two main exposures. Of the 272 participants screened and surveyed, the mean age was 11.2 (SD = 4.1). Majority of participants (n= 221, 81.6%) were male, while 50 (18.5%) were females. About half of the participants (n=133, 49.1%) got their household water from an unprotected source. Only 39 (14.3%) reported participating in farming as a domestic occupation.

Analytic Sample

The values which were coded as "missing" during variable preparation were excluded from the final analytic sample by the SAS Genmod procedure. The final sample size from which effects were estimated was 242 participants. (Figure 1)

Final adjusted model

At an alpha level of 0.05, the interaction term was found to be insignificant and dropped from the model.

Ln (probability of *S. haematobium*) = $\alpha + \beta 1$ (Water Contact) + $\beta 2$ (Sanitation) + $\gamma 1$ (Age) + $\gamma 2$ (Gender) + $\gamma 3$ (Farming) + $\gamma 4$ (Source of Water)

The general formulas used for estimating effects were:

- PR for the effect of water contact on *S. haematobium* prevalence: $\exp(\beta 1)$
- PR for the effect of sanitation on *S. haematobium* prevalence: $\exp(\beta 2)$
- PR for the effect of both exposures on *S. haematobium* prevalence: $\exp(\beta 1 + \beta 2)$

Estimated Effects of Individual Exposures

Tables 4 and 5 show the observed measures of association (prevalence ratios) between individual exposures - water contact and sanitation type – and the prevalence of *S. haematobium*. The number of outcome and summary measures for each effect group is also reported in the tables. Table 4 contains crude effects, while

the effects in table 5 have been adjusted for age, gender, water source, and domestic farming practices.

For the crude effects, participants who had moderate contact with water bodies were 3.03 times more likely to be infected, 95%CI [1.81, 5.07]. Those who had a high intensity of water contact were 2.41 times more likely to be infected, 95%CI [1.35, 4.32]. However, there was no statistical difference in the association when comparing the levels of water contact (High vs Moderate) with *S. haematobium* prevalence, PR = 0.80, 95 CI [0.55, 1.16]. Also, those with unimproved sanitation were equally likely to be infected with *S. haematobium* compared to those with improved sanitation, PR = 0.93, 95 CI [0.65, 1.35]. When considering the effect of any intensity of water contact alone, these participants were 7.32 times more likely to be infected with *S. haematobium*, 95%CI [2.61, 20.54].

For the adjusted estimates, moderate contact with water bodies was 2.14 times more likely to be associated with prevalence compared with no water contact, 95%CI [1.41, 3.23]. A high intensity of water contact was1.84 times more likely to be associated with prevalence, 95%CI [1.12, 3.03]. Again, there was no statistical difference in the association when comparing the levels of water contact (High vs Moderate) with *S. haematobium* prevalence, PR = 0.86, 95 CI [0.62, 1.21]. And those with unimproved sanitation were equally likely to be infected with *S. haematobium* compared to those with improved sanitation, PR = 0.99, 95 CI [0.72, 1.37]. Regardless of intensity, contact with contaminated water alone was associated with a 3.94 times increased likelihood of infection, 95%CI [1.68, 9.25].

Estimated Effects of Joint Exposure

The crude and adjusted effects of being exposed to both contaminated water and unimproved sanitation on *S. haematobium* prevalence compared to being exposed to neither are also highlighted in tables 4 and 5 respectively. As noted for individual effects, there was no statistical difference in the association when comparing high degree of water contact to some degree of water contact. Therefore, the overall combined effect of both exposures represented by any degree of water contact and the use of unimproved sanitation was also estimated.

For the crude prevalence ratios (table 4), the association of both moderate water contact and unimproved sanitation on prevalence compared to neither was 2.83 times more, 95%CI [1.54, 5.21]. The effect of high intensity of water contact and unimproved sanitation was 2.25, 95%CI [1.22, 4.17]. Overall, having any degree of water contact and unimproved sanitation had 6.83 times increased association with *S. haematobium* prevalence compared with having neither exposure, 95%CI [2.42, 19.30].

For the adjusted prevalence ratios (table 5), the association of both moderate water contact and unimproved sanitation on prevalence compared to neither was 2.11 times more, 95%CI [1.29, 3.48]. The effect of high intensity of water contact and unimproved sanitation was 1.83, 95%CI [1.10, 3.05]. And overall, having any degree of water contact and unimproved sanitation had 3.91 times increased association with *S. haematobium* prevalence compared with neither exposure, 95%CI [1.68, 9.10].

ANALYTIC SAMPLE



Figure 1. The final sample size utilized in the multivariate modeling.

Table 1. Prevalence of S. haematobium infection in this Study Population

S. haematobium Infection	N (%). Total = 272
Present	102 (37.5)
Absent	170 (62.50)

Intensity of Water Contact	Water Activity Scale (0 – 6)*	N (%). Total = 242
None	None	81 (33.5)
Some / Moderate	1 – 3 activities	100 (41.3)
High	4 – 6 activities	61 (25.2)

Table 2.1. Distribution of water contact activities reported by the participants

*Water activities assessed: playing, swimming, fetching, fishing, laundry, & working

Type of Sanitation	N (%). Total = 265
Unimproved	80 (30.2)
Improved	185 (69.8)

Table 2.2. Distribution of Sanitation type that participants have access to

		Intensity of Water Contact / Exposure		Sanita	tion Type	
	Total (272)	None	Some/Moderate	High	Improved	Unimproved
Characteristic	N (%) or Mean (SD)	N (%) or Mean (SD)	N (%) or Mean (SD)	N (%) or Mean (SD)	N (%) or Mean (SD)	N (%) or Mean (SD)
<i>S. haematobium</i> Infection						
Present Absent	102 (37.5) 170 (62.5)	14 (15.4) 67 (44.4)	52 (57.1) 48 (31.8)	25 (27.5) 36 (23.8)	67 (66.3) 118 (72.0)	34 (33.7) 46 (28.1)
Participant' age	11.2 (4.1)	9.0 (4.1)	11.9 (3.9)	12.9 (3.4)	11.2 (4.3)	11.2 (3.6)
Participant' gender						
Male Female	221 (81.6) 50 (18.5)	51 (26.6) 29 (59.2)	84 (43.8) 16 (32.7)	57 (29.7) 4 (8.2)	144 (67.3) 40 (80.0)	70 (32.7) 10 (20.0)
Source of Water						
Unprotected protected	133 (49.1) 138 (50.9)	31 (26.3) 50 (40.3)	51 (43.2) 49 (39.5)	36 (30.5) 25 (20.2)	85 (66.9) 100 (72.5)	42 (33.1) 38 (27.5)
Farming Occupation						
Farm Do not farm	39 (14.3) 233 (85.7)	6 (16.2) 75 (36.6)	22 (59.5) 78 (38.1)	9 (24.3) 52 (25.4)	34 (87.2) 151 (66.8)	5 (12.8) 75 (33.2)

Table 3. Participant Characteristics by degree of water contact and type of sanitation

Exposures			Number of <i>S.</i>		
Intensity of Water Contact	Type of sanitation	N	haematobium cases	Crude PR (95% CI)*	P value†
None	Improved	71	11	Ref. (1.00)	
Moderate	Improved	80	43	3.03 (1.81, 5.07)	<0.0001
High	Improved	31	13	2.41 (1.35, 4.32)	0.0030
High vs Moderate	Improved	131	65	0.80 (0.55, 1.16)	0.2301
None	Unimproved	10	3	0.93 (0.65, 1.35)	0.7139
Moderate	Unimproved	20	9	2.83 (1.54, 5.21)	0.0008
High	Unimproved	28	11	2.25 (1.22, 4.17)	0.0097
Any (Moderate or High)	Improved	131	65	7.32 (2.61, 20.54)	0.0002
Any (Moderate or High)	Unimproved	128	63	6.83 (2.42, 19.30)	0.0003

Table 4. Table of crude effects. Individual and joint associations between exposures and *S. haematobium* prevalence

Note: Crude model contained both exposures (water contact and sanitation) with no other covariates of interest.

*PR: Prevalence Ratios

†Absolute p values generated by the statistical software (SAS 9.4)

Exposures			Number of <i>S.</i>		
Intensity of Water Contact	Type of sanitation	N	haematobium cases	Adjusted PR (95% CI)*	P value
None	Improved	71	11	Ref. (1.00)	
Moderate	Improved	80	43	2.14 (1.41, 3.23)	0.0003
High	Improved	31	13	1.84 (1.12, 3.03)	0.0161
High vs Moderate	Improved	131	65	0.86 (0.62, 1.21)	0.3912
None	Unimproved	10	3	0.99 (0.72, 1.37)	0.9593
Moderate	Unimproved	20	9	2.11 (1.29, 3.48)	0.0031
High	Unimproved	28	11	1.83 (1.10, 3.05)	0.0205
Any (Moderate or High)	Improved	131	65	3.94 (1.68, 9.25)	0.0016
Any (Moderate or High)	Unimproved	128	63	3.91 (1.68, 9.10)	0.0016

Table 5. Table of adjusted effects. Individual and joint associations between exposures and S. haematobium prevalence

Note: Adjusted model contained all covariates: exposures and confounding factors.

*Adjusted for participant' age, gender, source of water for domestic use, and engagement in farming activities. †Absolute p values generated by statistical software (SAS 9.4)

V. DISCUSSION

Key Findings

This thesis examined data from a pilot cross-sectional study that was carried out among school-aged children in Kano Nigeria to estimate the individual and joint effects of water contact and sanitation practices on *S. haematobium* prevalence.

The results of the analyses suggest that water contact *at any degree or intensity* is not only significantly associated with *S. haematobium* prevalence, but also singularly confers the highest magnitude of association with the infection for this population (aPR = 3.94, 95% CI [1.68, 9.25]). Compared to improved sanitation, unimproved sanitation by itself is not significantly associated with prevalence (aPR = 0.99, 95% CI [0.72, 1.37]). Lastly, unimproved sanitation in combination with *any degree of water contact* is significantly associated with *S. haematobium* prevalence with an effect size of 3.91times the likelihood of infection (95% CI [1.68, 9.10]).

Joint Association

This thesis hypothesized that the joint effect of two key constructs, water contact and poor sanitation, which are embedded in the transmission of *S. haematobium* would be significantly higher than the effect size of either singular construct. However, this hypothesis was not confirmed. Although the joint effect of both exposures was significant (aPR = 3.91, 95% CI [1.68, 9.10]), the association remained slightly lower compared to the lone effect of water contact after adjusting for participants' age, gender, farming practices, and water source (aPR = 3.94, 95% CI [1.68, 9.25]).

Comprehensive reviews of previous literature document that the limiting factor for schistosomiasis transmission may differ from setting to setting. They elucidate that while contact with contaminated water is undeniably important, sanitation might play a more pivotal role in some settings and so, figuring out the limiting factor specific to a setting is pertinent for adequate control (Grimes et al., 2014).

The hypothesis tested here considered that regions still endemic for the disease are typically areas lacking in both 'safe' water and 'adequate' sanitation. Also, regions that have achieved significant control or disease eradication are notably those in which significant developmental changes occurred over time, including WASH improvements (Rollinson et al., 2013). Therefore, for WASH-poor Kano, Nigeria, focusing on one limiting factor over another might be counterproductive for significant and sustainable control. This thesis, therefore, sought to generate evidence that might highlight a stronger combined effect of multiple exposures compared to either individual exposure.

Perhaps, the finding of a weaker joint association compared to the effect of water contact alone does not deter from depicting the interwoven connections between the individual Water, Sanitation and Hygiene (WASH) components of schistosomiasis transmission. Indeed, it might serve to underscore this complexity. For instance, investigations into pathophysiologic mechanisms of infection chronicity report that bladder changes result in increased urgency and frequency of micturition. Those infected describe a consequent desire to urinate into water bodies due to the cooling effect of water on the genitals. (Rudge et al., 2008). In this, the co-occurrence of water contamination and poor sanitation behavior can be appreciated within the complex construct of water contact and possibly lends credence still to the call for integrated measures in the battle against schistosomiasis transmission (Akinwale et al., 2011; Gray et al., 2010).

WASH interventions have shown that the provision of safe water sources, while somewhat impactful, has not been synonymous with adequate prevention of contact with contaminated water. Also, the sole provision of hygiene education or sanitation facilities might only cause negligible effects in *S. haematobium* prevalence (Aagaard-Hansen et al., 2009; Grimes et al., 2015; Knopp et al., 2013). So, the reasoning is that the provision of accessible, safe, and functional sanitation facilities should reduce water contamination even during inevitable water contact. However, this needs to be backed by the provision of safe water sources and behavioral interventions that reduce contact with contaminated water (J. R. Stothard et al., 2009).

While this analyses could not generate absolute novel evidence to prove the reasoning above, the intertwined complexion of WASH exposures is still quite apparent and might need to be investigated and measured in a more direct or causal replication of this pilot.

Contact with Contaminated Water

This study was able to reiterate the established consensus that contact with contaminated water remains a principal determinant of schistosomiasis infection (WHO, 2019a). In analyzing the individual exposures for this setting, water contact was the sole limiting factor. It is apparent from underlying infection mechanisms that this is the most direct pathway in schistosome transmission; in that the infective schistosome cercariae that are excreted into water bodies by intermediate host snails have to penetrate the human skin when it comes into contact with the water (Colley et al., 2014).

In analyzing the association between water contact and prevalence, a scale denoting the degree of water contact was created from six water activities to preclude shrinking the construct of water contact in its quantification. Categories of high and moderate water contact intensities were created from this scale and used in effect estimation. Interestingly, this led to an additional finding that the prevalence of urinary schistosomiasis in this population statistically remained the same regardless of how many water contact activities the participants reportedly engaged in. This might support a strong correlation between the individual water contact activities, in that they aren't necessarily mutually exclusive (Afiukwa et al., 2019; Angelo et al., 2018). It also conveys the possibility that any contact with contaminated water puts individuals at similar levels of risk compared to a cumulative exposure index; thereby portraying a high infection potential of water contact behavior. With this finding, the need for more causal estimation of risk, that incorporates frequency of water contact, for instance, becomes apparent (Grimes et al., 2015).

Also, the collection, preparation, and analysis of "water contact" as a construct support suggestions that the underlying determinants of this behavior might be complex. This ambiguity is repeatedly reported in the evaluation of various waterbased interventions that have aimed at reducing *S. haematobium* prevalence. Interventions have differed and underlying indicators of impact measurements have also differed (Aagaard-Hansen et al., 2009). For example, some have provided piped or bore-hole water as a safe water alternative for domestic use and irrigation at the community level. With this, a secondary mitigation of water contact is usually anticipated, but sustainable impact is highly dependent on inherent behavioral practices and how readily available or accessible water sources are (Peter Steinmann, Jennifer Keiser, Robert Bos, Marcel Tanner, & Jürg Utzinger, 2006). A few interventions have directly provided alternate sources of water recreation to detract from continued water contamination and contact. However, the impacts have largely hinged on hygiene behavior and / or functional sanitation infrastructure (Evan Secor, 2014; Kosinski et al., 2012). These lessons need to be incorporated in designing any intervention to reduce or eliminate contact with contaminated water.

Sanitation

In estimating the individual role that sanitation plays on *S.haematobium* prevalence for this population, the measured index was the use of unimproved sanitation compared to improved sanitation facilities and this demonstrated no significant association with prevalence. Notably, preliminary descriptive analysis of this data had shown that a high proportion (70%) of participants infected with *S. haematobium* reported the use of improved sanitation facilities as opposed to unimproved facilities. This is a remarkable finding since endemic regions such as

Kano, Nigeria are also typically deficient in adequate sanitation (Liao et al., 2011) (Akpabio, 2012) (Kullmann et al., 2018).

Relatively few studies have been carried out and able to demonstrate lower odds of infection with improved sanitation (Grimes et al., 2014). In particular, those that have collected adequate data to demonstrate that those with improved sanitation had the facilities in their homes (Knopp et al., 2013). Overall, however, investigations into the role that sanitation plays has majorly involved evaluating combined sanitation and hygiene interventions (Rollinson et al., 2013).

Peradventure, the insignificant finding here does more to support the complexity of measuring sanitation as a construct and in the context of schistosomiasis transmission (Grimes et al., 2015). Reviews of many schistosomiasis related interventions have found it difficult to estimate the role of sanitation due to differing indicators. In Nigeria, for instance, the availability of improved sanitation infrastructure has repeatedly unequaled its use because said infrastructure might be inaccessible or nonfunctional (Asaolu & Ofoezie, 2003).

Furthermore, the construct of sanitation might be embedded within the measurement of "water contact" wherein initial water contamination occurs. There is an increasing need to define "use" and "access" as regards sanitation infrastructure. The latter could include safety, privacy, facility cleanliness, distance to facility, functionality during the rainy season, shared access, and so on (Hulland et al., 2015). There are also calls for more nuanced approaches in collecting information about

sanitation due to interviewer biases and stigma that could be related to inadequate facilities (WHO, 2019b).

Strengths and Limitations

Context is highly important and informs relevant data-driven strategy for schistosomiasis interventions. The factors influencing prevalence have been known to vary within regions of a country (Clements et al., 2009). This study exhibited strength in highlighting the roles played by the investigated exposures on disease for the specific context of Kano Nigeria. The simple random sampling strategy employed reduced bias; electronic coding and collection minimized data entry errors, and the population of school-aged children used is suitable for inference comparable to other studies. Albeit a small study, it was able to generate transparent and directly applicable evidence that highlights the interwoven intricacies of the WASH factors on schistosomiasis transmission.

However, due to several limitations, caution should be employed in interpreting the findings of this study. This pilot could only cover five Local Government Areas (LGAs). Furthermore, the simple random sampling strategy employed would be limited in the generalizability of findings to the entire region of Kano. For better inference, complex sampling methods would be more representative – weighted based on previous prevalence estimates for the different LGAs. Self reported answers to the survey questions are liable to recall and interviewer biases. Also, the cross-sectional and observational nature of this study limits adequate measurements of exposures and causal inference from results. Ffinally, there is an apparent weakness in the way data on sanitation was collected. The questionnaire assessing type of sanitation facility did not account for customary use of, actual access to, and functionality of sanitation infrastructure. In addition, a pertinent assessment of sanitation behavior for this peculiar water-borne disease, such as the occurrence and frequency in which participants urinated into water bodies was not collected (Grimes et al., 2015). Future research would benefit from the rigorous application of principles and guidelines for the collection of WASH information (Gine, Jiménez Fdez de Palencia, & Pérez-Foguet, 2013).

Implications for Practice and Research

The findings of this study emphasize the significant role of water contact in the prevalence of urinary schistosomiasis for the Kano population of Nigeria. It particularly goes further to portray that this exposure - disease association is possibly the key limiting factor for this setting when compared to poor sanitation. Although the estimated role of joint water contact and sanitation exposures was slightly lower than the singular role of water contact, its significant effect size is not hastily discarded as it presents a novel leverage for further research that could support the need for integrated WASH research and interventions in the battle against schistosomiasis.

Nigeria, the country with the most cases of infection, is in dire need of more relevant data to inform a targeted national strategy (Ezeh et al., 2019). Without a doubt, highly endemic regions such as in Kano, are lacking across all WASH indices (Akpabio, 2012). For these data and in this setting, although water contact was demonstrated as a limiting factor and sanitation was not, we interpret these observational findings with care and recognize that while solutions to poor sanitation alone may not be pivotal for control, sanitation behavior is still embedded in water contact. Robust research is therefore necessary to investigate if disease transmission would likely continue if both constructs are not attended to concurrently or if adequately designed programs targeting water contact alone might end up killing two birds with one stone (Rollinson, 2009).

In conclusion, this pilot does not only begin to highlight the complex interrelations between individual WASH components in *Schistosoma haematobium* infection. It also provides direction for more focused causal research to estimate risk and control strategies that consider the combined roles of WASH exposures.

REFERENCES

- Aagaard-Hansen, J., Mwanga, J. R., & Bruun, B. (2009). Social science perspectives on schistosomiasis control in Africa: past trends and future directions. *Parasitology*, 136(13), 1747-1758. doi:10.1017/S0031182009006404
- Abdulkadir, A., Ahmed, M., Abubakar, B. M., Suleiman, I. E., Yusuf, I., Imam, I. M., . . . Musa, B. M. (2017). Prevalence of urinary schistosomiasis in Nigeria, 1994–2015: Systematic review and meta-analysis. *African Journal of Urology*, 23(3), 228-239. doi:<u>https://doi.org/10.1016/j.afju.2016.11.004</u>
- Abdullahi, M., Bassey, S., & Oyeyi, T. (2009). A Comprehensive Mapping of Urinary Schistosomiasis Using Geographic Information Systems (GIS) in Kano State, Nigeria. *Bayero Journal of Pure ans Applied Sciences, 2*(1). doi:http://dx.doi.org/10.4314/bajopas.v2i1.73453
- Abe, E. M., Guan, W., Guo, Y.-H., Kassegne, K., Qin, Z.-Q., Xu, J., . . . Zhou, X.-N. (2018). Differentiating snail intermediate hosts of Schistosoma spp. using molecular approaches: fundamental to successful integrated control mechanism in Africa. *Infectious diseases of poverty, 7*(1), 29-29. doi:10.1186/s40249-018-0401-z
- Adedini, S. A., Odimegwu, C., Bamiwuye, O., Fadeyibi, O., & De Wet, N. (2014). Barriers to accessing health care in Nigeria: implications for child survival. *Global health action*, 7, 23499-23499. doi:10.3402/gha.v7.23499
- Adenowo, A. F., Oyinloye, B. E., Ogunyinka, B. I., & Kappo, A. P. (2015). Impact of human schistosomiasis in sub-Saharan Africa. *The Brazilian Journal of Infectious Diseases*, 19(2), 196-205. doi:<u>https://doi.org/10.1016/j.bjid.2014.11.004</u>
- Afiukwa, F. N., Nwele, D. E., Uguru, O. E., Ibiam, G. A., Onwe, C. S., Ikpo, A. U., . . . Odoemena, O. F. (2019). Transmission Dynamics of Urogenital Schistosomiasis in the Rural Community of Ebonyi State, South Eastern Nigeria. *Journal of Parasitology Research*, 1 8. doi:<u>https://doi.org/10.1155/2019/7596069</u>
- Akinwale, O. P., Kane, R. A., Rollinson, D., Stothard, J. R., Ajayi, M. B., Akande, D. O., . . . Adeleke, M. A. (2011). Molecular approaches to the identification of Bulinus species in southwest Nigeria and observations on natural snail infections with schistosomes. *Journal of Helminthology*, 85(3), 283-293. doi:10.1017/S0022149X10000568
- Akpabio, E. M. (2012) Water supply and sanitation services sector in Nigeria: The policy trend and practice constraints. In, *ZEF Working Paper Series: Vol. 96*: Zentrum für Entwicklungsforschung / Center for Development Research(ZEF), University of Bonn.
- Amuta, E. U., & Houmsou, R. S. (2014). Prevalence, intensity of infection and risk factors of urinary schistosomiasis in pre-school and school aged children in Guma Local Government Area, Nigeria. Asian Pacific Journal of Tropical Medicine, 7(1), 34-39. doi:<u>https://doi.org/10.1016/S1995-7645(13)60188-1</u>
- Angelo, T., Buza, J., Kinung'hi, S. M., Kariuki, H. C., Mwanga, J. R., Munisi, D. Z., & Wilson, S. (2018). Geographical and behavioral risks associated with Schistosoma haematobium infection in an area of complex transmission. *Parasites & Vectors*, 11(1), 481. doi:10.1186/s13071-018-3064-5

- Asaolu, S. O., & Ofoezie, I. E. (2003). The role of health education and sanitation in the control of helminth infections. *Acta Tropica*, *86*(2), 283-294. doi:<u>https://doi.org/10.1016/S0001-706X(03)00060-3</u>
- Atalabi, T. E., Adoh, S. D., & Eze, K. M. (2018). The current epidemiological status of urogenital schistosomiasis among primary school pupils in Katsina State, Nigeria: An imperative for a scale up of water and sanitation initiative and mass administration of medicines with Praziquantel. *PLoS neglected tropical diseases, 12*(7), e0006636-e0006636. doi:10.1371/journal.pntd.0006636
- Bartram, J., & Cairncross, S. (2010). Hygiene, sanitation, and water: forgotten foundations of health. *PLoS medicine*, 7(11), e1000367-e1000367. doi:10.1371/journal.pmed.1000367
- Bassey, S., & Umar, Z. (2004). A re-assessment of Schistosomiasis infection in Garun-Babba, Kadawa and Kura in Kano State, Nigeria. *Nigerian Journal of Parasitology*, 25(1), 107-109. doi:<u>http://dx.doi.org/10.4314/njpar.v25i1.37716</u>
- Bergquist, R. (2013). Closing in on 'perhaps the most dreadful of the remaining plagues': an independent view of the multidisciplinary alliance to optimize schistosomiasis control in Africa. Acta Tropica, 128(2), 179-181. doi:https://doi.org/10.1016/j.actatropica.2013.08.016
- Braun, L., Grimes, J. E. T., & Templeton, M. R. (2018). The effectiveness of water treatment processes against schistosome cercariae: A systematic review. *PLoS neglected tropical diseases*, *12*(4), e0006364-e0006364. doi:10.1371/journal.pntd.0006364
- Bruun, B., & Aagaard-Hansen, J. (2008). The Social Context of Schistosomiasis and its Control. An Introduction and Annotated Bibliography. Geneva: UNICEF/UNDP/World Bank/WHO Special Programme for Research & Training in Tropical Diseases (TDR). Retrieved from
- Center for Global Health . Division of Parasitic, D., & Malaria. (2011). The Targeted neglected tropical diseases (NTDs). 1. Retrieved from <u>https://stacks.cdc.gov/view/cdc/12083</u>
- Chiamah, O., Ubachukwu, P., Anorue, C., & Ebi, S. (2019). Urinary schistosomiasis in Ebonyi State, Nigeria from 2006 to 2017. *Journal of Vector Borne Diseases, 56*(2), 87-91. doi:10.4103/0972-9062.263721
- Clements, A. C. A., Firth, S., Dembelé, R., Garba, A., Touré, S., Sacko, M., . . . Fenwick, A. (2009). Use of Bayesian geostatistical prediction to estimate local variations in Schistosoma haematobium infection in western Africa. *Bulletin of the World Health Organization*, 87(12), 921-929. doi:10.2471/BLT.08.058933
- Clennon, J. A., Mungai, P. L., Muchiri, E. M., King, C. H., & Kitron, U. (2006). SPATIAL AND TEMPORAL VARIATIONS IN LOCAL TRANSMISSION OF SCHISTOSOMA HAEMATOBIUM IN MSAMBWENI, KENYA. *The American Journal of Tropical Medicine and Hygiene*, *75*(6), 1034-1041. doi:https://doi.org/10.4269/ajtmh.2006.75.1034
- Colley, D. G., Bustinduy, A. L., Secor, W. E., & King, C. H. (2014). Human schistosomiasis. *Lancet.* 383(9936):2253-2264., 383(9936). Retrieved from https://stacks.cdc.gov/view/cdc/36773
- Colley, D. G., & Secor, W. E. (2007). A Schistosomiasis Research Agenda. *PLoS neglected tropical diseases*, 1(3), e32. doi:10.1371/journal.pntd.0000032
- Dawaki, S., Al-Mekhlafi, H. M., Ithoi, I., Ibrahim, J., Abdulsalam, A. M., Ahmed, A., . . . Surin, J. (2016). PREVALENCE AND RISK FACTORS OF SCHISTOSOMIASIS AMONG HAUSA COMMUNITIES IN KANO STATE, NIGERIA. *Revista do Instituto de Medicina Tropical de Sao Paulo, 58*, 54-54. doi:10.1590/S1678-9946201658054
- Dawaki, S., Al-Mekhlafi, H. M., Ithoi, I., Ibrahim, J., Abdulsalam, A. M., Ahmed, A., . . . Atroosh, W. M. (2015). The Menace of Schistosomiasis in Nigeria: Knowledge, Attitude, and

Practices Regarding Schistosomiasis among Rural Communities in Kano State. *PLOS ONE,* 10(11), e0143667. doi:10.1371/journal.pone.0143667

- Development, D. f. I. (2012). *London declaration on neglected tropical diseases*. Retrieved from <u>http://www.dfif.gov.uk</u>
- Doenhoff, M. J., Hagan, P., Cioli, D., Southgate, V., Pica-Mattoccia, L., Botros, S., . . . Engels, D. (2009). Praziquantel: Its use in control of schistosomiasis in sub-Saharan Africa and current research needs. *Parasitology*, *136*(13), 1825-1835. doi:10.1017/S0031182009000493
- Donohue, R. E., Mashoto, K. O., Mubyazi, G. M., Madon, S., Malecela, M. N., & Michael, E. (2017). Biosocial Determinants of Persistent Schistosomiasis among Schoolchildren in Tanzania despite Repeated Treatment. *Tropical medicine and infectious disease*, 2(4), 61. doi:10.3390/tropicalmed2040061
- Ekanem, E., Akapan, F., & Eyong, M. (2017). Urinary schistosomiasis in school children of a southern nigerian community 8 years after the provision of potable water. *Nigerian Postgraduate Medical Journal*, 24(4), 201-204. doi:10.4103/npmj.npmj_136_17
- Espino, F. E., Koops, V., Manderson, L., Research, U. N. W. B. W. S. P. f., & Training in Tropical, D. (2004). Community participation and tropical disease control in resource-poor settings / Fe Espino, Vaughn Koops, Lenore Manderson. In. Geneva: World Health Organization.
- Evan Secor, W. (2014). Water-based interventions for schistosomiasis control. *Pathogens and global health*, *108*(5), 246-254. doi:10.1179/2047773214Y.0000000149
- Eyong, M., Ikpeneme, E., & Ekanem, E. (2008). Relationship between Schistosoma haematobium infection and urinary tract infection among children in South Eastern, Nigeria. *The Nigerian Postgraduate Medical Journal*, *15*(2), 89-93.
- Ezeh, C. O., Onyekwelu, K. C., Akinwale, O. P., Shan, L., & Wei, H. (2019). Urinary schistosomiasis in Nigeria: a 50 year review of prevalence, distribution and disease burden. [La schistosomiase urinaire au Nigeria : un bilan de 50 ans sur la prévalence, la distribution et le fardeau de la maladie]. *Parasite (Paris, France), 26*, 19-19. doi:10.1051/parasite/2019020
- Folefac, L. N., Nde-Fon, P., Verla, V. S., Tangye, M. N., Njunda, A. L., & Luma, H. N. (2018).
 Knowledge, attitudes and practices regarding urinary schistosomiasis among adults in the Ekombe Bonji Health Area, Cameroon. *The Pan African medical journal, 29*, 161-161. doi:10.11604/pamj.2018.29.161.14980
- Freeman, M. C., Ogden, S., Jacobson, J., Abbott, D., Addiss, D. G., Amnie, A. G., . . . Utzinger, J. (2013). Integration of Water, Sanitation, and Hygiene for the Prevention and Control of Neglected Tropical Diseases: A Rationale for Inter-Sectoral Collaboration. *PLoS Negl Trop Dis. 2013; 7(9). 7(9).* Retrieved from https://stacks.cdc.gov/view/cdc/21163
- French, M., & Evans, D. (2019). Adaptive strategies for schistosomiasis control. *The Lancet Global Health, 7*(10). doi:https://doi.org/10.1016/S2214-109X(19)30371-7
- French, M. D., Evans, D., Fleming, F. M., Secor, W. E., Biritwum, N.-K., Brooker, S. J., . . . Zhang, Y. (2018). Schistosomiasis in Africa: Improving strategies for long-term and sustainable morbidity control. *PLoS neglected tropical diseases, 12*(6), e0006484-e0006484. doi:10.1371/journal.pntd.0006484
- Garba, A., Lamine, M. S., Barkiré, N., Djibo, A., Sofo, B., Gouvras, A. N., . . . Utzinger, J. (2013).
 Efficacy and safety of two closely spaced doses of praziquantel against Schistosoma haematobium and S. mansoni and re-infection patterns in school-aged children in Niger.
 Acta Tropica, 128(2), 334-344. doi:https://doi.org/10.1016/j.actatropica.2012.08.008

- Gine, R., Jiménez Fdez de Palencia, A., & Pérez-Foguet, A. (2013). Water-sanitation-hygiene mapping: An improved approach for data collection at local level. *The Science of the total environment, 463-464C*, 700-711. doi:10.1016/j.scitotenv.2013.06.005
- Gray, D. J., McManus, D. P., Li, Y., Williams, G. M., Bergquist, R., & Ross, A. G. (2010). Schistosomiasis elimination: lessons from the past guide the future. *The Lancet Infectious Diseases*, 10(10), 733-736. doi:<u>https://doi.org/10.1016/S1473-3099(10)70099-2</u>
- Grimes, J. E. T., Croll, D., Harrison, W. E., Utzinger, J., Freeman, M. C., & Templeton, M. R. (2014). The Relationship between Water, Sanitation and Schistosomiasis: A Systematic Review and Meta-analysis. *PLoS neglected tropical diseases*, 8(12), e3296. doi:10.1371/journal.pntd.0003296
- Grimes, J. E. T., Croll, D., Harrison, W. E., Utzinger, J., Freeman, M. C., & Templeton, M. R. (2015). The roles of water, sanitation and hygiene in reducing schistosomiasis: a review. *Parasites & Vectors*, 8(1), 156. doi:10.1186/s13071-015-0766-9
- Gryseels, B., Polman, K., Clerinx, J., & Kestens, L. (2006). Human schistosomiasis. *The Lancet,* 368(9541), 1106-1118. doi:<u>https://doi.org/10.1016/S0140-6736(06)69440-3</u>
- Hajissa, K., Muhajir, A. E. M. A., Eshag, H. A., Alfadel, A., Nahied, E., Dahab, R., . . . Mohamed, Z. (2018). Prevalence of schistosomiasis and associated risk factors among school children in Um-Asher Area, Khartoum, Sudan. *BMC research notes*, *11*(1), 779-779. doi:10.1186/s13104-018-3871-y
- Holveck, J. C., Ehrenberg, J. P., Ault, S. K., Rojas, R., Vasquez, J., Cerqueira, M. T., . . . Periago, M. R. (2007). Prevention, control, and elimination of neglected diseases in the Americas: Pathways to integrated, inter-programmatic, inter-sectoral action for health and development. *BMC Public Health*, 7(1), 6. doi:10.1186/1471-2458-7-6
- Hotez, P. J., Engels, D., Fenwick, A., & Savioli, L. (2010). Africa is desperate for praziquantel. *The Lancet*, *376*(9740), 496-498. doi:10.1016/S0140-6736(10)60879-3
- Hulland, K. R. S., Chase, R. P., Caruso, B. A., Swain, R., Biswal, B., Sahoo, K. C., . . . Dreibelbis, R. (2015). Sanitation, Stress, and Life Stage: A Systematic Data Collection Study among Women in Odisha, India. *PLOS ONE, 10*(11), e0141883-e0141883. doi:10.1371/journal.pone.0141883
- King, C. H. (2009). Global health: Toward the elimination of schistosomiasis. *New England Journal of Medicine*, *360*(2), 106-109. doi:10.1056/NEJMp0808041
- King, C. H. (2017). The evolving schistosomiasis agenda 2007-2017-Why we are moving beyond morbidity control toward elimination of transmission. *PLoS neglected tropical diseases*, 11(4), e0005517-e0005517. doi:10.1371/journal.pntd.0005517
- King, C. H., Sturrock, R. F., Kariuki, H. C., & Hamburger, J. (2006). Transmission control for schistosomiasis - why it matters now. *Trends in Parasitology*, 22(12), 575-582. doi:10.1016/j.pt.2006.09.006
- Kjetland, E., Ndhlovu, P., Gomo, E., Mduluza, T., Midzi, N., Gwanzura, L., . . . Gundersen, S. (February 2006). Association between genital schistosomiasis and HIV in rural Zimbabwean women. *AIDS, 20*(4), 593 600.
- Knopp, S., Stothard, J. R., Rollinson, D., Mohammed, K. A., Khamis, I. S., Marti, H., & Utzinger, J. (2013). From morbidity control to transmission control: time to change tactics against helminths on Unguja Island, Zanzibar. *Acta Tropica*, *128*(2), 412-422. doi:<u>https://doi.org/10.1016/j.actatropica.2011.04.010</u>
- Kosinski, K. C., Adjei, M. N., Bosompem, K. M., Crocker, J. J., Durant, J. L., Osabutey, D., . . . Gute, D. M. (2012). Effective control of Schistosoma haematobium infection in a Ghanaian

community following installation of a water recreation area. *PLoS neglected tropical diseases, 6*(7), e1709-e1709. doi:10.1371/journal.pntd.0001709

- Kullmann, C. P., Andres, L. A., Srivastava, V., Skoufias, E., Li, S., Loughnan, E. C., ... Rana, S. (2018). Reducing inequalities in water supply, sanitation, and hygiene in the era of the sustainable development goals synthesis report of the WASH poverty diagnostic initiative : Reducing Inequalities in Water Supply, Sanitation, and Hygiene in the Era of the Sustainable Development Goals (English). Retrieved from Washington, D.C.: http://documents.worldbank.org/curated/en/554021528950823000/Reducing-Inequalities-in-Water-Supply-Sanitation-and-Hygiene-in-the-Era-of-the-Sustainable-Development-Goals
- Lansdown, R., Ledward, A., Hall, A., Issae, W., Yona, E., Matulu, J., . . . Bundy, D. (2002). Schistosomiasis, helminth infection and health education in Tanzania: Achieving behaviour change in primary schools. *Health Education Research*, *17*(4), 425-433. Retrieved from <u>https://www.scopus.com/inward/record.uri?eid=2-s2.0-</u> 0036346011&partnerID=40&md5=e623cbd4c059b8313ef34e2165aabb10
- Liao, C. W., Sukati, H., Nara, T., Tsubouchi, A., Chou, C. M., & Jian, J. Y. (2011). Prevalence of Schistosoma haematobium infection among schoolchildren in remote areas devoid of sanitation in northwestern Swaziland, Southern Africa. *Jpn J Infect Dis., 64*.
- McManus, D. P., Dunne, D. W., Sacko, M., Utzinger, J., Vennervald, B. J., & Zhou, X.-N. (2018). Schistosomiasis. *Nature Reviews Disease Primers*, 4(1), 13. doi:10.1038/s41572-018-0013-8
- Moe, C. (2009). Achieving Water and Sanitation Services for Health in Developing Countries. In Institute of Medicine (US) Roundtable on Environmental Health Sciences, Research, and Medicine. Global Environmental Health: Research Gaps and Barriers for Providing Sustainable Water, Sanitation, and Hygiene Services: Workshop Summary. Retrieved from https://www.ncbi.nlm.nih.gov/books/NBK50770/
- Muhammad, F., Abdulkareem, J. H., & Chowdhury, A. A. (2017). Major Public Health problems in Nigeria. *South East Asia Journal Of Public Health, 7*, 6-11. doi:<u>http://dx.doi.org/10.3329/seajph.v7i1.34672</u>
- Mwanga, J. R., & Lwambo, N. J. S. (2013). Pre- and post-intervention perceptions and water contact behaviour related to schistosomiasis in north-western Tanzania. Acta Tropica, 128(2), 391-398. doi:<u>https://doi.org/10.1016/j.actatropica.2012.09.017</u>
- N'Goran, E. K., Utzinger, J., N'Guessan, A. N., Müller, I., Zamblé, K., & Lohourignon, K. L. (2001). Reinfection with Schistosoma haematobium following school-based chemotherapy with praziquantel in four highly endemic villages in Côte d'Ivoire. *Trop Med Int Health., 6*. doi:10.1046/j.1365-3156.2001.00785.x
- Nmorsi, O., Kwandu, U., & Ebiaguanye, L. (2007). Schistosoma haematobium and urinary tract pathogens co-infections in a rural community of Edo State, Nigeria. *The Journal of Communicable Diseases*, *39*(2), 85-90.
- Ojurongbe, O., Sina-Agbaje, O. R., Busari, A., Okorie, P. N., Ojurongbe, T. A., & Akindele, A. A. (2014). Efficacy of praziquantel in the treatment of Schistosoma haematobium infection among school-age children in rural communities of Abeokuta, Nigeria. *Infectious diseases of poverty, 3*, 30-30. doi:10.1186/2049-9957-3-30
- Onyebuchi, C. (2012). *Nigeria mater plan for neglected tropical disease (NTDs) 2013-2017*. Retrieved from <u>http://docplayer.net/2404435-Nigeria-master-plan-for-neglected-tropical-diseases-ntds-2013-2017.html</u>

Osakunor, D. N. M., Woolhouse, M. E. J., & Mutapi, F. (2018). Paediatric schistosomiasis: What we know and what we need to know. *PLoS neglected tropical diseases, 12*(2), e0006144-e0006144. doi:10.1371/journal.pntd.0006144

Portal, N. D. (2020). Nigeria Census. Retrieved from <u>https://nigeria.opendataforafrica.org/xspplpb/nigeria-census#</u>

- Rollinson, D. (2009). A wake up call for urinary schistosomiasis: Reconciling research effort with public health importance. *Parasitology*, *136*(12), 1593-1610. doi:10.1017/S0031182009990552
- Rollinson, D., Knopp, S., Levitz, S., Stothard, J. R., Tchuem Tchuenté, L.-A., Garba, A., . . . Utzinger, J. (2013). Time to set the agenda for schistosomiasis elimination. *Acta Tropica*, 128(2), 423-440. doi:<u>https://doi.org/10.1016/j.actatropica.2012.04.013</u>
- Rosenthal, P. J. (2020). Schistosomiasis (Bilharziasis). In M. A. Papadakis, S. J. McPhee, & M. W. Rabow (Eds.), *Current Medical Diagnosis and Treatment 2020*. Retrieved from accessmedicine.mhmedical.com/content.aspx?aid=1166619853
- Rudge, J. W., Stothard, J. R., Basáñez, M.-G., Mgeni, A. F., Khamis, I. S., Khamis, A. N., & Rollinson, D. (2008). Micro-epidemiology of urinary schistosomiasis in Zanzibar: Local risk factors associated with distribution of infections among schoolchildren and relevance for control. *Acta Tropica*, 105(1), 45-54. doi:https://doi.org/10.1016/j.actatropica.2007.09.006
- Secor, W. E., & Colley, D. G. (2018). When Should the Emphasis on Schistosomiasis Control Move to Elimination? *Tropical medicine and infectious disease*, 3(3), 85. doi:10.3390/tropicalmed3030085
- Singh, K., Muddasiru, D., & Singh, J. (2016). Current status of schistosomiasis in Sokoto, Nigeria. *Parasite epidemiology and control, 1*(3), 239-244. doi:10.1016/j.parepi.2016.08.003
- Sokolow, S. H., Wood, C. L., Jones, I. J., Swartz, S. J., Lopez, M., Hsieh, M. H., . . . De Leo, G. A. (2016). Global Assessment of Schistosomiasis Control Over the Past Century Shows Targeting the Snail Intermediate Host Works Best. *PLoS neglected tropical diseases*, 10(7), e0004794. doi:10.1371/journal.pntd.0004794
- Steinmann, P., Keiser, J., Bos, R., Tanner, M., & Utzinger, J. (2006). Schistosomiasis and water resources development: systematic review, meta-analysis, and estimates of people at risk. *Lancet Infect Dis.*, 6. doi:10.1016/S1473-3099(06)70521-7
- Steinmann, P., Keiser, J., Bos, R., Tanner, M., & Utzinger, J. (2006). Schistosomiasis and water resources development: systematic review, meta-analysis, and estimates of people at risk. *The Lancet Infectious Diseases*, 6(7), 411-425. doi:<u>https://doi.org/10.1016/S1473-3099(06)70521-7</u>
- Stothard, J. R., Chitsulo, L., Kristensen, T. K., & Utzinger, J. (2009). Control of schistosomiasis in sub-Saharan Africa: progress made, new opportunities and remaining challenges. *Parasitology*, 136(13), 1665-1675. doi:10.1017/S0031182009991272
- Stothard, J. R., French, M. D., Khamis, I. S., Basáñez, M.-G., & Rollinson, D. (2009). The epidemiology and control of urinary schistosomiasis and soil-transmitted helminthiasis in schoolchildren on Unguja Island, Zanzibar. *Transactions of The Royal Society of Tropical Medicine and Hygiene*, 103(10), 1031-1044. doi:<u>https://doi.org/10.1016/j.trstmh.2009.03.024</u>
- Tanser, F., Azongo, D. K., Vandormael, A., Bärnighausen, T., & Appleton, C. (2018). Impact of the scale-up of piped water on urogenital schistosomiasis infection in rural South Africa. *eLife*, 7, e33065. doi:10.7554/eLife.33065
- Tchuem Tchuenté, L. A., Southgate, V. R., Webster, B. L., De Bont, J., & Vercruysse, J. (2001). Impact of installation of a water pump on schistosomiasis transmission in a focus in

Cameroon. *Transactions of The Royal Society of Tropical Medicine and Hygiene, 95*(3), 255-256. doi:10.1016/S0035-9203(01)90228-7

- Thomassen Morgas, D. E., Kvalsvig, J. D., Taylor, M., Gundersen, S. G., & Kjetland, E. F. (2010). Schistosomiasis and water-related practices in school girls in rural KwaZulu-Natal, South Africa. Southern African Journal of Epidemiology and Infection, 25(4), 30-33. doi:10.1080/10158782.2010.11441406
- Ugbomoiko, U. S., Ofoezie, I. E., Okoye, I. C., & Heukelbach, J. (2010). Factors associated with urinary schistosomiasis in two peri-urban communities in south-western Nigeria. *Annals* of Tropical Medicine and Parasitology, 104(5), 409-419. doi:10.1179/136485910X12743554760469
- Umar, S., Shinkafi, S. H., Hudu, S. A., Neela, V., Suresh, K., Nordin, S. A., & Malina, O. (2017).
 Prevalence and molecular characterisation of Schistosoma haematobium among primary school children in Kebbi State, Nigeria. *Annals of Parasitology, 63*(2), 133-139. doi:<u>https://doi.org/10.17420/ap6302.97</u>
- Utzinger, J., N'Goran, E. K., Caffrey, C. R., & Keiser, J. (2011). From innovation to application: Social–ecological context, diagnostics, drugs and integrated control of schistosomiasis. *Acta Tropica*, *120*, S121-S137. doi:<u>https://doi.org/10.1016/j.actatropica.2010.08.020</u>
- Utzinger, J., Raso, G., Brooker, S., De Savigny, D., Tanner, M., ØRnbjerg, N., ... N'Goran, E. K. (2009). Schistosomiasis and neglected tropical diseases: towards integrated and sustainable control and a word of caution. *Parasitology*, *136*(13), 1859-1874. doi:10.1017/S0031182009991600
- Wang, L., Utzinger, J., & Zhou, X.-N. (2008). Schistosomiasis control: experiences and lessons from China. *The Lancet, 372*(9652), 1793-1795. doi:<u>https://doi.org/10.1016/S0140-6736(08)61358-6</u>
- WHO. (1991). Basic Laboratory methods in medical parasitology. In *Urine filtration SOP* (pp. 41-44). Geneva: World Health Organization.
- WHO. (2013). *Schistosomiasis: progress report 2001–2011, strategic plan 2012–2020*. Retrieved from Geneva, Switzerland:
- WHO. (2019a). Schistosomiasis. *Newsletters*. Retrieved from <u>https://www.who.int/news-room/fact-sheets/detail/schistosomiasis</u>
- WHO. (2019b). Surveillance of water, sanitation and hygiene in schools. A practical tool;
 Copenhagen: ; . Licence: CC BY-NC-SA 3.0 IGO. In. Copenhagen: WHO Regional Office for Europe.
- WHO/UNICEF. (2019, 2019). WASH MONITORING INDICES SANITATION. Retrieved from https://washdata.org/monitoring/sanitation
- Winters, K., & Netscher, S. (2016). Proposed Standards for Variable Harmonization Documentation and Referencing: A Case Study Using QuickCharmStats 1.1. PLOS ONE, 11(2), e0147795. doi:10.1371/journal.pone.0147795
- Yapi, Y. G., Briët, O. J. T., Diabate, S., Vounatsou, P., Akodo, E., Tanner, M., & Teuscher, T. (2005).
 Rice irrigation and schistosomiasis in savannah and forest areas of Côte d'Ivoire. Acta Tropica, 93(2), 201-211. doi:<u>https://doi.org/10.1016/j.actatropica.2004.11.005</u>
- Zida, A., Briegel, J., Kabré, I., Sawadogo, M. P., Sangaré, I., Bamba, S., . . . Wacker, J. (2016).
 Epidemiological and clinical aspects of urogenital schistosomiasis in women, in Burkina
 Faso, West Africa. *Infectious diseases of poverty*, 5(1), 81. doi:10.1186/s40249-016-0174-1